Author's response to reviews

Title: Paget's Disease of the Breast in a Male with Lymphomatoid Papulosis.

Authors:

Dina Fouad (dina238@googlemail.com)

Version: 4 Date: 7 July 2010

Author's response to reviews: see over
07 July 2010

Journal of Medical Case Reports Editorial Team

Dear Sir or Madam,

**Subject: Revisions of Article**

**Article title: Paget’s Disease of the Breast in a Male with Lymphomatoid Papulosis.**

**MS ID : 1400972461356082**

I enclosed a revised version of my article and have highlighted and tracked the sentence which I believe may have been missing. The oldest version of my article I have is dated the 24th October 2009 and there do not seem to be any differing sections in the remainder of my article versions. If there is still believed to be a portion missing that was previously present I would greatly appreciate if I could please be sent this version and I will quickly complete the necessary amendments.

Thank you.

Yours Sincerely,

Dina Fouad